Author's response to reviews

Title: The combination of ANT2 shRNA and hNIS radioiodine gene therapy increases CTL cytotoxic activity through the phenotypic modulation of cancer cells

Authors:

Yun Choi (choi@snu.ac.kr)
Ho Won Lee (lee5@empal.com)
Jaetae Lee (jtlee@empas.com)
Yong Hyun Jeon (jeon9014@gmail.com)

Version: 6 Date: 15 January 2013

Author's response to reviews: see over
Dear Editor in BMC cancer

We are pleased to resubmit our manuscript titled “The combination of ANT2 shRNA and hNIS radioiodine gene therapy increases CTL cytotoxic activity through the phenotypic modulation of cancer cells”. To answer the reviewer's comments in Abstract section, we further described the methods and results for the level of caspase-3 activation in treated cells in Abstract section. We highlighted all changes made when revising the manuscript with yellow color.

We sincerely hope that this revised manuscript is acceptable for publication in BMC Cancer.

Best and kindest regards,

Yong Hyun Jeon, Ph.D.
Author's response to comments

Title: The combination of ANT2 shRNA and hNIS radioiodine gene therapy increases CTL cytotoxic activity through the phenotypic modulation of cancer cells

Authors: Yun Choi, Ho Won Lee, Jaetae Lee, Yong Hyun Jeon

Reviewer: Dongmei He

Version: 5 Date: 6 January 2013

(Reviewer's report)

I feel the manuscript's publication could now go ahead. Minor Essential Revisions: The assay of the level of caspase-3 activation (additional experiment) should be shown in the methods and results of the abstract.

(Answer)

We appreciate reviewer's comments. To faithfully answer the reviewer's comment, we further described the methods and results for the level of caspase-3 activation in treated cells in Abstract section. Please find revised abstract section.